Author’s response to reviews

Title: Detection of domestic violence by community mental health teams: a multi-center, cluster randomized controlled trial

Authors:
Roos Ruijne (r.ruijne@erasusmc.nl)
Louise Howard (louise.howard@kcl.ac.uk)
Kylee Trevillion (kylee.trevillion@kcl.ac.uk)
Femke Jongejan (f.jongejan@erasusmc.nl)
Carlo Garofalo (C.Garogalo@uvt.nl)
Stefan Bogaerts (S.Bogaerts@uvt.nl)
Niels Mulder (c.l.mulder@erasusmc.nl)
Astrid Kamperman (a.kamperman@erasusmc.nl)

Version: 1 Date: 20 Jun 2017

Author’s response to reviews:

Roos Ruijne, MD
Erasmus Medical Center
‘s Gravendijkwal 230
3015 CE Rotterdam, the Netherlands
r.ruijne@erasusmc.nl

Subject: Rebuttal BPSY-D-17-00447 'Detection of domestic violence by community mental health teams: a multi-center, cluster randomized controlled trial’

Date 20-06-2017
Dear editor,

On behalf of all editors I would like to convey our appreciation for the evaluation of our manuscript submitted to BMC Psychiatry: BPSY-D-17-00447 'Detection of domestic violence by community mental health teams: a multi-center, cluster randomized controlled trial’. We will address the comments below.

1. Ethical and Funding Approval Documentation

Before we can proceed with your submission, can you please forward copies of all ethical approval and funding approval for our records. These documents should be sent as email attachments to the following email address, BMCSeriesEditorial@biomedcentral.com. Please DO NOT upload these documents as additional files in the submission system. If your documents are not in English, please could you provide translated versions of the relevant parts. These should be endorsed and signed by a contactable person at the institution. Please also include the original documents. Can you please confirm in your cover letter that you have forwarded the requested documentation to BMCSeriesEditorial@biomedcentral.com.

I hereby confirm that a copy of the ethical and funding approval and a translation of the relevant parts have been send to BMCSeriesEditorial@biomedcentral.com on 20-06-2017.

2. Funding

This study is funded by the Netherlands Organisation for Scientific Research. This is a governmental institution which funds research in health. The study proposal was peer reviewed by the Netherlands Organization for Scientific Research.

3. Study status

This is an ongoing study

4. Related Articles

Data from this study have not been submitted, published or presented.
5. Trial registration

The registration date (15/01/2015) has been added to the abstract.

6. Ethics approval and consent to participate

A consent to participate statement is added to this section. Participants will be asked for written consent to participate in the study.

We look forward to receiving your positive reply.

Best regards,

Roos Ruijne

MD, PhD Candidate BRAVE study